Author's response to reviews

Title: Synergistic effect of acyclovir and 3,19-isopropylideneandrographolide on herpes simplex virus wild types and drug resistant strains

Authors:

Thongkoon Priengprom (thong.koontoy2@hotmail.com)
Tipaya Ekalaksananan (tipeka@kku.ac.th)
Bunkerd Kongyingyoes (bunkon@kku.ac.th)
Supawadee Suebsasana (hnungnet@yahoo.com)
Chantana Aromdee (chaaro@kku.ac.th)
Chamsai Pientong (chapie@kku.ac.th)

Version: 2
Date: 13 August 2014

Author's response to reviews:

***EDITORIAL REQUIREMENTS***

The line and page numbers have been added in the revised manuscript.
The page breaks are removed.